Author's response to reviews

Title: Angle-torque relationship of the subtalar pronators and supinators in younger and elderly males and females

Authors:

Marco Hagen MH (marco.hagen@uni-due.de)
Daniel Sanchez-Bergmann DSB (daniel@sanchez-bergmann.de)
Sebastian Seidel SS (seidel-man@web.de)
Matthias Lahner ML (m.lahner@klinikum-bochum.de)

Version: 4 Date: 16 November 2015

Author's response to reviews:

Dear editor,

1. References were included in a single square bracket throughout the manuscript.
2. "Insert table/figure here" sections were removed.
3. Unnecessary capitals were removed.
4. References were removed from the conclusion.
5. In the online submission system, the authors' suffixes (MH, DSB, SS, ML) were removed from the author details.

Kind regards,
Marco Hagen and Matthias Lahner